Author’s response to reviews

Title: Increased Risk of Hip Fractures in Patients with Newly Diagnosed Dementia: A Nationwide Population-based Study

Authors:

Hao-Kuang Wang (ed101393@edah.org.tw)
Yi-Cheng Tai (ed106560@edah.org.tw)
Kang Lu (kanglu.1002@gmail.com)
Po-Chou Liliang (ed100172@edah.org.tw)
Chi-Wei Lin (ed104283@edah.org.tw)
Yi-Che Lee (pipijer@gmail.com)
Pei-Hsuan Fang (phoebe.phtang@gmail.com)
Li-Ching Chang (changlc@isu.edu.tw)
Ying-Chun Li (ycli@faculty.nsysu.edu.tw)

Version: 2 Date: 15 May 2014

Author’s response to reviews:

Dear Editor,

We revised the manuscript format based on the instructions. Please see the revised manuscript. Thank you!

Sincerely,
Ying-Chun Li